VC- /Week 3 starting   15Jan07
SM18 Operation Meeting (Tuesdays) 16 Jan  07
HOW THE SITUATION WAS AT END 2006 and  at START UP ON 8 Jan 2007
	2
	A1
	MBAR2411
	PT 13.2 Resist. Meas.
	295.18

	8
	A2
	MBAR2268
	ICS 1 WP04 HV Test Warm
	294.19

	9
	B1
	SSS519
	PT 14 Magnet MAPS
	294.28

	1
	B2
	SSS516
	ICS 5 Traveller Last
	295

	7
	C1
	SSS099
	Prep 5 Cool Down
	161.17

	5
	C2
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	294.55

	3
	D1
	SSS669
	PT 13.2 Resist. Meas.
	294.55

	10
	D2
	SSS509
	Prep 0 Scan Bench
	295

	4
	E1
	MBAR2402
	PT 13.2 Resist. Meas.
	295.44

	6
	E2
	MBBR2394
	Prep 5 Cool Down
	274.22

	11
	F1
	MBBL2197
	Prep 2.3 Res. Meas. CDW1
	294.01

	12
	F2
	MBAR2407
	ICS 4 Final connection
	293.74


1/8/2007 4:56:34 PM   Tempr. from TBSM18 site

E2 & C1 were leak checked already before Xmas shutdown
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VC- /Week 3 starting   15Jan07
SM18 Operation Meeting (Tuesdays) 16 Jan  07

Bench status  on monday 15-Jan-2007 at   1/15/2007 2:54:50 PM
	7
	A1
	MBBR2384
	Prep 5 Cool Down
	173.84

	5
	A2
	MBAR2268
	PTE Min. NRJ 3KA Diode Test
	13.88

	8
	B1
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	295.62

	9
	B2
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	296.34

	1
	C1
	SSS099
	PT 13.2 Resist. Meas.
	295.18

	10
	C2
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	295.62

	11
	D1
	SSS523
	ICS 3 Traveller after warm test
	295.8

	6
	D2
	SSS509
	PT 6.2 Training 2
	1.90

	12
	E1
	MBAR2414
	ICS 1 WP04 HV Test Warm
	296.52

	3
	E2
	MBBR2394
	PT 13.2 Resist. Meas.
	296.34

	4
	F1
	MBBL2197
	PT 13.2 Resist. Meas.
	294.64

	2
	F2
	MBAR2407
	ICS 6 Disconnect Magnet WP07
	295.36


CONFIG :  D 6 - AS 2 -  IRSSS 4 ;   
Supervisor:  Arjan ;        Eq support 16-3602 : 

4 ( + 2 ) Magnet Tests completed ( 1 Magnet Stripped ) ; 

2 Dipoles tests


1 Arc-SSS (  370 )

0 500-series ( )

1  600-series [ 669 – last one : was completed in Dec]
+ B cluster magnets 516, 519  already done  were NOT removed till Monday 15th 
 SSS099 + 2394 were tested but s99 had ROBTX & 2394 had HV problems at CDBP

 Situation from Cryogenics:  J-P Lamboy .  
General Points:

Work Hours : ICS & Cryo Teams   ( Eric + Kevin)

Control Room Lock up  - pin code  ( Eric) 

2 -person Rule
· Situation  all SSS  : J Halik : 
· Situation Cryo ( J-P Lamboy) & Feb working
· A.O.B
Tests 

	Magnets under Test from 07-01-2007 to 13-01-2007 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAR2411 
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	13-12-2006 
	61 h.
	30 h.
	100 h.
	8 h.
	470 h.
	10-01-2007 
	669 h.
	Stripping
	
	

	2
	TBC2
	SSS370 
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	17-12-2006 
	0 h.
	29 h.
	10 h.
	9 h.
	52 h.
	08-01-2007 
	100 h.
	Stripping
	
	

	3
	TBD1
	SSS669 
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	14-12-2006 
	37 h.
	26 h.
	53 h.
	10 h.
	495 h.
	09-01-2007 
	621 h.
	Stand By
	
	

	4
	TBE1
	MBAR2402 
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	16-12-2006 
	42 h.
	21 h.
	15 h.
	17 h.
	552 h.
	12-01-2007 
	647 h.
	Stand By
	
	


In addition, SSS099 tested twice ( ROBTC) ; 2394 had HV problems at cold & was warmed up

 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR2411
	10704.4 A.
(D2_U)
	11688 A.
(D1_U)
	12514.8 A.
(D1_U)
	12520 A.
(D1_L)
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	-
	-
	-
	-
	-
	-
	-
	-

	2
	SSS370
	11991.2 A.
(QE_12)
	12190.4 A.
(QE_12)
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS669
	3513 A.
(Ext_MQML)
	3998.6 A.
(Ext_MQML)
	4157 A.
(Ext_MQM)
	4367.1 A.
(Int_MQM)
	4355 A.
(Ext_MQM)
	4447.4 A.
(Int_MQML)
	4483.4 A.
(E_MQM)
	4499 A.
(I_MQM)
	4505 A.
(I_MQML)
	4529.5 A.
(Ex_MQML)
	4574.9 A.
(Ex_MQML)
	4587.1 A.
(Int_MQM)
	4602.8 A.
(Ext_MQML)
	4638.5 A.
(Int_Mqml)
	4650 A.
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	4
	MBAR2402
	12229.6 A.
(D2_L)
	12615 A.
(D2_U)
	12850 A.
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	


MTM TEAM  (6:00-22:00)

Shift Leader 

160060

OP1




160483

OP2




160077

Nick




160957

ICS TEAM (7:30-16:00)

ICS (Chef de banc)

162926

ROCLA




165059

JOSEPH ROMANELLO
164649
CRYO TEAM (6:00-22:00)
CRYO MOBILE

160354
EXPERT (OLD Equipment support) (Normal hours)
HV

Richard MOMPO
164157


MM

Guy DEFERNE
164284

IAP/HF/LF: 
Giorgio D’ANGELO 163793

Salem BOUCHENOUA 163557

ANTI-CRYOSTAT:Patrick Legrand1632
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	08/01/2007
	TO DO LIST for SHIFT 

	 
	 
	START MORNING SHIFT

	SMS
	1
	New elogbook session.

	
	2
	KEY CLUSTER A,B,C,D,E,F present on electrical box. Then SM18CR key.

	
	3
	Mobile present Shift leader 160060, OP1:160077, OP2:160483

	
	4
	Check priority and consigned on with board, then previous logbook session.

	
	5
	Check Any problem from cryo during night

	
	X
	Start test on bench (If still are ready)

	 
	 
	END MORNING SHIFT

	EMS
	X
	Test on bench (If still are going on).

	
	1
	Put a maximum of information in elogbook for next shift.

	
	2
	Pass information to next shift.

	 
	 
	START AFTERNOON SHIFT

	SAS
	1
	Received information from previous shift.

	
	2
	New elogbook session.

	
	X
	Test on bench (If still are going on).

	 
	 
	END AFTERNOON SHIFT

	EAS
	X
	STOP test on bench (If still are going on).

	
	1
	Put a maximum of information in elogbook for next shift.

	
	2
	KEY CLUSTER A,B,C,D,E,F present on electrical box. Then SM18CR key.

	
	3
	Mobile present Shift leader 160060, OP1:160077, OP2:160483

	
	4
	Check clearty of priority and consigned on with board.

	
	5
	Request Cryo operator to close the SM18CR. And put Key in cupboard.

	
	
	

	
	 
	New elogbook session

	
	 
	Check Equipment

	
	 
	Test on bench

	
	 
	Information exchange with CRYO or Previous/Next Shift.
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TERMINATION COUNTER -2007

Remaining of the totally 1250 Dipoles cryostated  that still  need testing this year

	No
	MAGNET
	BENCH
	ARRIVAL
	DEPARTURE
	Comments

	1
	2407
	F2
	19-12-2006
	
	

	2
	2268
	A2
	21-12-2006
	
	

	3
	2384
	A1
	10-01-2007
	
	

	4
	2394
	E2
	16-12-2006
	
	

	5
	2326 (MM)
	
	
	
	

	6
	2263
	
	
	
	

	7
	2267 (MM)
	
	
	
	

	8
	2399(MM)
	
	
	
	

	9
	2423
	
	
	
	

	10
	2424
	
	
	
	

	11
	2425
	
	
	
	

	12
	2426
	
	
	
	

	13
	2324
	
	
	
	

	14
	2551
	
	
	
	

	15
	2598
	
	
	
	

	16
	2739
	
	
	
	

	17
	2328
	
	
	
	

	18
	2790
	
	
	
	

	19
	2624
	
	
	
	

	20
	2414
	E1
	12-12-2006
	
	

	21
	2408
	
	
	
	

	22
	2420
	
	
	
	

	23
	2419
	
	
	
	

	24
	2418
	
	
	
	

	25
	2326
	
	
	
	

	26
	1334
	
	
	
	


15 of  26 Dipoles for completion of Tunnel                            Updated : January 16, 2007
TERMINATION COUNTER -2007    [ ALL SSS ]

	29
	SSS 509
	
	10-01-2007
	
	

	30
	SSS 523
	
	10-01-2007
	
	

	31
	SSS 516
	
	15-12-2006
	15-01-07
	Done

	32
	SSS 529
	
	
	
	

	33
	SSS 530
	
	
	
	

	34
	SSS 531
	
	
	
	

	35
	SSS 370
	
	12 DEC 06

17 Dec :ROBTC
	
	

	36
	SSS 361
	
	
	
	

	37
	SSS 099
	
	18/12/2006
	
	

	38
	SSS 374
	
	
	
	

	39
	SSS 356
	
	
	
	

	40
	SSS 367
	
	
	
	

	41
	SSS 369
	
	
	
	

	42
	SSS 364
	
	
	
	


Updated : January 16, 2007
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_09_A
	TBE2
	MB2394: HV test at cold failed dipole vs ground
	MTM
	
	Richard contacted. IFS box removed for investigation. Fault confirmed. Magnet warm up naturally and removed without test.

	Tuesday_09_A
	TBC1
	SSS099: Powering of corrector not possible: Power converter was not going to standby.
	MTM
	
	Giorgio contacted. He found the switch on wrong position (off position).

	Tuesday_09_A
	TBD2
	SSS509: During HV initial @ warm MQTL vs Ground failed.
	MTM
	HV
	Richard investigated the problem: capacity limit in the software corrected. Solved.

	Tuesday_09_A
	TBA2
	MB2268: Leak in the magnet found by cryo team found during capillary leak test.
	ICS
	
	ICS team changed the join in the line E.

	Monday_15_M
	TBD2
	SSS509: Training 2 could not be launched: PLCSE blocked.
	MTM
	
	TEMA and Power Converter reset but still fault. Giorgio informed. He did the same manipulation three times and then OK.



































































